# Decoding the Nexus Between Credit Risk Challenges and Banks Performance:

# Perspectives from Commercial Banks in Jordan

Odai AlJaloudi<sup>1</sup>, Nicholas Dacre<sup>2</sup>, M.K.S. Al-Mhdawi<sup>3</sup>,

<sup>1</sup>University of Gloucestershire, UK; <sup>2</sup>University of Southampton, UK; <sup>3</sup>Teesside University, UK;

#### **Abstract**

With the increasing complexity of global financial markets, commercial banks in emerging markets face a multitude of challenges in managing credit risk, which directly impacts their performance. While the importance of credit risk management is widely recognised, comprehensive research on the connection between credit risk challenges and banks' performance, particularly in emerging markets, remains scarce. Existing literature on credit risk management often focuses on developed markets, neglecting the unique challenges faced by banks in emerging markets. As such, there is a growing need for empirical studies to explore the specific credit risk challenges encountered by commercial banks in emerging markets and their potential implications for banks' performance. This study aims to address this gap by investigating the relationship between credit risk challenges and banks' performance in emerging markets, focusing on the case of Jordan. The research adopts a literature review and administers a questionnaire to explore and assess key credit risk challenges and their implications for performance. Preliminary analysis highlights critical issues such as cybersecurity threats, fluctuations in interest rates, inadequate resources and expertise, changes in regulatory frameworks, and the lack of system integration within the banks' internal departments. This research contributes to a deeper understanding of the credit risk management landscape in emerging markets and provides valuable insights for policymakers, regulators, and banking practitioners. Addressing these challenges can enable commercial banks to enhance their credit risk management practices, improve performance, and ultimately contribute to the stability and resilience of the financial system in emerging markets.

**Keywords:** Emerging markets, credit risk, banks, performance, cybersecurity, interest rates, regulation, resources, integration, Jordan.

# 1. Introduction

The global financial landscape has become increasingly complex, presenting commercial banks with significant challenges, particularly in the area of credit risk management (AlJaloudi et al., 2024; Bi & Bao, 2024; Chen et al., 2021; Dacre, AlJaloudi, Al-Mhdawi, et al., 2024; Dacre, AlJaloudi, Thiam, et al., 2024; Saliba et al., 2023). Credit risk, defined as the potential for financial loss arising from a borrower's failure to repay a loan or fulfil contractual obligations, is a critical factor influencing banks' performance and stability (Ghosh, 2015). Recent events underscore the importance of effective credit risk management (Eggleton et al., 2023; Kou et al., 2021; Sonjit et al., 2021a). For example, in 2023, several banks including Silicon Valley Bank and Signature Bank, collapsed due to significant liquidity issues and poorly managed credit exposures, leading to substantial financial losses and market instability (Feinstein et al., 2024; Metrick, 2024; Ozili, 2024). Similarly, the failure of First Republic Bank in 2023 highlighted the consequences of inadequate risk assessment and overexposure to high-risk assets (Nobanee et al., 2023), while the collapse of Silvergate Bank, primarily serving cryptocurrency clients, highlighted weaknesses associated with concentrated exposure to volatile sectors (Warren, 2024). These instances illustrate a critical need for effective credit risk management practices in order to maintain financial stability (Chen et al., 2021; Nobanee et al., 2023; Ozili, 2024; Saliba et al., 2023).

While the importance of credit risk management is universal, its manifestation and management differ significantly between developed and emerging markets (Dacre, AlJaloudi, Al-Mhdawi, et al., 2024; Dacre, AlJaloudi, Thiam, et al., 2024; Hopper et al., 2017; Hoskisson et al., 2000; Levine, 1996). For instance, in developed economies such as the United States and the European Union, established regulatory frameworks and advanced credit scoring systems provide instruments to mitigate risks (Anderson, 2007; Claessens et al., 2024; Miller, 2003). However despite these frameworks, challenges do persist, for instance as experienced through the European sovereign debt crisis, where high-risk loans and limited oversight strained banks (De Marco, 2019; Lane, 2012; Mitchener & Trebesch, 2023). In contrast, emerging markets face distinct obstacles such as limited access to credit histories, less established institutional frameworks, and heightened exposure to external disruptions (Dacre, AlJaloudi, Al-Mhdawi, et al., 2024; Kane, 1995; Zalan & Toufaily, 2017). The case of India's Non-Banking Financial Companies (NBFCs)

crisis in 2018, driven by high defaults and inadequate governance, highlights the acute nature of these challenges in less mature financial systems (Manda & Rani, 2019).

More specifically, emerging markets are typically distinguished by rapid economic growth, financial market liberalisation, and evolving regulatory environments (Bekaert & Harvey, 2003; Dong et al., 2021b; Saeed, 2024; Singh, 1997; Stiglitz, 2003). These characteristics tend to introduce unique challenges for commercial banks, such as inadequate infrastructure and limited access to reliable data (AlJaloudi et al., 2024; Costa et al., 2014; Hawkins & Mihaljek, 2001; Mohsin et al., 2023; Simpson, 2002; Vong & Song, 2015). Consequently, credit risk management practices that prove effective in developed markets may not always seamlessly adapt to the complex and dynamic contexts of emerging economies (Dacre, AlJaloudi, Al-Mhdawi, et al., 2024; Mateey, Sahyouni, et al., 2023). Examining these challenges is essential in order to develop appropriate risk management strategies suited to these environments. However despite this mandate, the existing body of research largely focuses on credit risk management in developed markets, often overlooking the distinct challenges encountered by banks in emerging economies (Ayayi, 2012; Saleh & Abu Afifa, 2020; Van Greuning & Bratanovic, 2020). The omission is significant due to the growing integration and prominence of emerging markets in the global financial system and the potential for credit risk events in these regions to trigger broader economic consequences (AlJaloudi et al., 2024; Audi & Al-Masri, 2024; Bekaert & Harvey, 2003; Chari, 2023; Dacre, AlJaloudi, Thiam, et al., 2024; Prasad et al., 2003; Stiglitz, 2003).

Bridging this research gap is vital for several reasons. Firstly, understanding the specific credit risk challenges encountered by banks in emerging markets can inform the development of tailored risk management strategies and best practices that reflect the unique features of these markets (Bo, 2024; Dacre, AlJaloudi, Al-Mhdawi, et al., 2024; Scott et al., 2024). Secondly, identifying the potential impact of credit risk challenges on banks' performance can highlight areas where policy interventions and regulatory reforms may be required to support financial stability and economic growth (Abdou & Alarabi, 2024; Dacre, AlJaloudi, Al-Mhdawi, et al., 2024; Ezeoha, 2011; Karanja et al., 2022). Lastly, examining the relationship between credit risk challenges and banks' performance in emerging markets can contribute to the broader discourse on financial resilience and sustainability, guiding efforts to foster a more stable and inclusive global financial system

(Bekaert & Harvey, 2003; Costa et al., 2014; Dacre, AlJaloudi, Thiam, et al., 2024; Dacre, Yan, Frei, et al., 2024; Stiglitz, 2003; Vong & Song, 2015).

This study therefore seeks to bridge this gap by investigating the relationship between credit risk challenges and banks' performance in emerging markets, with a particular focus on Jordan (Al-Smadi & Al-Wabel, 2011; Dacre, AlJaloudi, Al-Mhdawi, et al., 2024; Mashhour & Zaatreh, 2008). As an emerging market economy, Jordan has witnessed significant growth and development in its banking sector over recent decades (Mazur, 2023; Moh'd AL-Tamimi et al., 2019). However, such as many other emerging markets, Jordanian banks face a number of credit risk challenges that could impair their performance and contribute to financial instability (Alzuod & Alqhaiwi, 2022; Barber et al., 2021; Kharabsheh & Gharaibeh, 2022; Kutum & Al-Jaberi, 2015; Sonjit et al., 2021b).

The remainder of this paper is structured as follows: Section 2 reviews the literature on credit risk management, highlighting its impact on banks' performance and the unique challenges in emerging markets, including economic vulnerabilities, resource constraints, regulatory issues, and evolving risks like cybersecurity and interest rate fluctuations. Section 3 outlines the research methodology, while section 4 discusses the initial findings, highlighting the key credit risk challenges identified and their potential implications for banks' performance. Finally, Section 5 concludes the paper by summarising the main insights, proposing avenues for future research, and offering recommendations for policymakers, regulators, and banking practitioners.

#### 2. Literature Review

Studies in developed markets have consistently highlighted the pivotal role of credit risk management in ensuring financial stability and optimising banks' performance. For instance, research has demonstrated a negative relationship between Non-Performing Loans (NPLs) and profitability, showing that higher NPL ratios weaken financial performance by reducing interest income and increasing provisioning requirements (Bayar, 2019; Berger & DeYoung, 1997; Chen et al., 2021; Louzis et al., 2012). Similarly, macroeconomic factors such as GDP growth, inflation, and unemployment have been identified as critical determinants of credit risk exposure, shaping the broader economic conditions under which banks operate (Awijen et al., 2023; Ghosh, 2015; Mpofu & Nikolaidou, 2018; Reinhart & Rogoff, 2011)

Table 1: Contrasting Credit Risk Challenges in Developed and Emerging Economies

Aspect	Developed Markets	Emerging Markets
Macroeconomic Factors	Stable economic conditions with manageable inflation and low unemployment rates (Awijen et al., 2023).	High macroeconomic volatility, inflationary pressures, and higher unemployment rates (Moussa, 2015).
Regulatory Frameworks	Well-established regulations (i.e. Basel III compliance) (Frait & Tomšík, 2014).	Developing regulatory frameworks, resource-constrained (Bekaert & Harvey, 2003).
Credit Information Systems	Advanced systems for credit scoring and borrower assessment (Anderson, 2007).	Developing credit information systems, limiting accurate borrower assessments (Siddique et al., 2022).
Technological Integration	High adoption of advanced fintech and analytics tools (Serkbayeva et al., 2024).	Limited adoption of fintech solutions due to resource and infrastructure constraints (Ng & Kwok, 2017).
Bank Size and Capitalisation	Larger banks with established capitalisation and risk management frameworks (Louzis et al., 2012).	Smaller banks with emergent capitalisation, more vulnerable to credit defaults (Das & Ghosh, 2007).
Exposure to External Shocks	Less susceptible to external shocks due to diversified economies (Bayar, 2019).	Susceptible to external shocks, such as currency volatility and global interest rate changes (Hawkins & Mihaljek, 2001).
Cybersecurity Threats	Funded systems to mitigate and respond to cyberattacks (Kamiya et al., 2021).	Legacy technological infrastructure increases vulnerability to cyber risks (Ani et al., 2017).
Access to Credit Risk Tools	Advanced risk assessment tools, supported by big data and predictive modelling (Ghosh, 2015).	Limited access to advanced tools and data, hindering risk assessment capabilities (Poudel, 2012).
Loan Growth Dynamics	Moderate loan growth, closely monitored to minimise defaults (Berger & DeYoung, 1997).	Rapid loan growth, often with limited risk monitoring (Das & Ghosh, 2007).
Focus on Sustainability	Growing focus on climate- related risks and ESG considerations (Serkbayeva et al., 2024)	Limited focus on sustainability due to pressing economic challenges (Beshouri & Gravråk, 2010).

While these insights provide valuable lessons, their applicability to emerging markets is limited (Dacre, AlJaloudi, Al-Mhdawi, et al., 2024). The economic, institutional, and regulatory conditions in emerging markets differ markedly from those in developed economies, introducing unique challenges (Hawkins & Mihaljek, 2001). Developing legal frameworks, underdeveloped financial systems, and heightened susceptibility to external shocks exacerbate credit risk in these regions (Bekaert & Harvey, 2003) (Table 1). For instance, studies from South Asia and North Africa highlight how macroeconomic

volatility and inadequate credit information systems amplify default risks in emerging banking sectors (Moussa, 2015; Siddique et al., 2022)

Although the literature on credit risk management in emerging markets is less developed, some studies address this gap. In India and Pakistan, for example, bank size, loan growth, and macroeconomic conditions have been identified as significant contributors to credit risk, with smaller banks and those experiencing rapid loan growth being particularly vulnerable to credit defaults (Das & Ghosh, 2007; Manda & Rani, 2019; Siddique et al., 2022). Research in Malaysia has similarly demonstrated the importance of effective credit risk management practices in mitigating adverse effects on financial performance, highlighting the need for effective internal controls and regulatory oversight (Al-Mhdawi, O'Connor, et al., 2023; Al-Mhdawi et al., 2024; Karim et al., 2010).

Comparative studies have shown that Jordanian banks lag behind their regional counterparts in terms of risk management efficiency, though these studies have not specifically addressed credit risk or its implications for bank performance (Al-Tamimi & Al-Mazrooei, 2007; AlJaloudi et al., 2024; Alshatti, 2015; Dacre, AlJaloudi, Thiam, et al., 2024). Findings suggest a significant relationship between credit risk management practices and the profitability of Jordanian commercial banks, but they underscore the need for further research to explore the specific challenges faced by this sector (Alshatti, 2015).

The COVID-19 pandemic (Eggleton et al., 2021; Sonjit et al., 2021c) has underscored the fragility of global credit systems, with emerging markets experiencing significant surges in non-performing loans. These challenges, combined with the increasing focus on climate-related risks (Dong et al., 2021a; Tite et al., 2021a), are reshaping the landscape of credit risk management, necessitating a more holistic and adaptive approach (Al-Mhdawi, Dacre, et al., 2023; Baxter et al., 2023; Dong et al., 2022; Dong et al., 2024). Additionally, emerging trends, such as the integration of fintech considerations into banking practices, are beginning to influence credit risk management (Dacre, AlJaloudi, Thiam, et al., 2024; Serkbayeva et al., 2024). For instance, the adoption of advanced analytics and risk assessment tools has been shown to enhance the accuracy of credit scoring models, particularly in emerging economies where traditional credit evaluation methods are often inadequate (Anderson, 2007; Ghosh, 2015; Serkbayeva et al., 2024). However, the digitalisation of financial services can present potential cybersecurity threats, adding a new

dimension of risk to the evolving landscape of credit risk management (Claessens et al., 2024; Gong et al., 2022; Zhang et al., 2023).

# 2.1 Cybersecurity Threats

Cybersecurity incidents, such as data breaches and cyberattacks, can result in significant financial losses, reputational harm, and erosion of customer trust (Kamiya et al., 2021; Mohamed et al., 2024) (Table 2).

Table 2: Notable Cybersecurity Breaches and Their Impact on Credit Risk Management

Incident	Year	Scope	Implications for Credit Risk Management	Source(s)
Bangladesh Bank Cyber Heist	2016	\$81 million stolen via fraudulent SWIFT transactions.	Demonstrated vulnerabilities in financial systems; disrupted banking operations; exposed systemic risks to credit systems reliant on secure interbank transactions.	(Bukth & Huda, 2017; Mazumder & Sobhan, 2020)
Equifax Data Breach	2017	Exposed sensitive information of ~147 million individuals, including Social Security numbers and credit histories.	Compromised credit assessments; eroded trust in credit reporting systems; led to financial losses and legal penalties.	(Kenny, 2018; Novak & Vilceanu, 2019; Zou et al., 2018)
Capital One Data Breach	2019	Exposed personal data of ~100 million customers, including credit scores and payment histories.	Disrupted lending operations; damaged customer trust; heightened the need for robust cybersecurity measures in credit risk management.	(Khan et al., 2022; Novaes Neto et al., 2020)
Colonial Pipeline Ransomware Attack	2021	Caused a major disruption in fuel supply and resulted in a \$4.4 million ransom payment.	Highlighted indirect risks to credit systems through operational disruptions; increased focus on operational resilience in risk management frameworks.	(Beerman et al., 2023; Dudley & Golden, 2021)
Optus Data Breach	2022	Compromised personal data, including passport numbers and Medicare IDs of 2.1 million individuals.	Raised concerns over the security of sensitive customer data; increased potential for fraud, identity theft, and inaccurate credit risk assessments.	(Analytica, 2022; Biddle et al., 2022)

Within the realm of credit risk management, cybersecurity breaches can compromise sensitive customer information, disrupt lending operations, and undermine the integrity of credit assessment processes (Eling et al., 2021; Schlackl et al., 2022). For example, the

2017 Equifax data breach exposed sensitive information of approximately 147 million individuals, including Social Security numbers and credit histories (Kenny, 2018; Zou et al., 2018). This breach not only caused substantial financial losses and legal penalties but also compromised the integrity of credit assessments (Novak & Vilceanu, 2019). Similarly, the 2019 Capital One breach affected over 100 million customers, exposing credit scores, payment histories, and other sensitive data, which disrupted lending operations and damaged customer trust (Khan et al., 2022; Novaes Neto et al., 2020). More recently, the 2022 Optus breach compromised personal information such as passport numbers and Medicare IDs of 2.1 million individuals, raising significant concerns about the security of customer data and the broader implications for credit risk management (Biddle et al., 2022).

From a credit risk management perspective, cybersecurity breaches present specific challenges (Grima et al., 2023; Scott et al., 2024). The theft of sensitive customer data, such as credit histories and financial statements can compromise the accuracy and reliability of credit evaluations (Abidin et al., 2019; Avery et al., 2004; Peretti, 2008). Disruptions to lending platforms and payment systems caused by cyberattacks can also lead to operational inefficiencies, delays, and errors in credit processing and monitoring (Youvan, 2024). These disruptions not only affect current operations but can also have long-term implications for risk assessment models and credit portfolios (Jorion, 2000).

Studies have also emphasised the growing prevalence of cybersecurity threats in the banking sector (Cele & Kwenda, 2024; Uddin et al., 2020). For instance, Aldasoro et al. (2020) analysed the frequency and severity of cyber incidents in financial institutions, finding that the number of reported incidents has increased markedly in recent years. The study also notes that the financial sector is particularly susceptible to cyber risks due to its reliance on complex, interconnected systems and the sensitive nature of the data it manages (Aldasoro et al., 2022). Kamiya et al. (2021) examined the effect of data breaches on banks' stock returns and found that the announcement of a data breach leads to a significant negative impact on stock prices (Tweneboah-Kodua et al., 2018). Findings suggests that the reputational damage and potential legal liabilities associated with data breaches can detrimentally affect banks' financial performance (Tosun, 2021).

In the context of emerging markets, cybersecurity challenges are particularly acute due to legacy technological infrastructure, which leaves systems susceptible to exploitation (Ani et al., 2017; McLennan, 2022). Financial constraints often result in insufficient investment in advanced cybersecurity tools and defences (Arcuri et al., 2017). Moreover, the rapid growth of internet penetration and mobile banking services can outpace the development of effective security measures, further intensifying vulnerabilities (Beshouri & Gravråk, 2010; Falaiye et al., 2024; Sultana, 2009). Legacy systems, compounded by a lack of staff training and awareness, amplify the risks (Huo et al., 2024). These issues, alongside the accelerated pace of digitalisation and increased reliance on fintech solutions, expose banks in these regions to significant cyber threats (Ng & Kwok, 2017). Employing structured frameworks such as the NIST Cybersecurity Framework, particularly its 'Detect' function, has been shown to enhance fraud prevention by facilitating real-time threat identification and response (Krumay et al., 2018; Shen, 2014). Strategies including multi-factor authentication, machine learning-driven transaction monitoring, and endpoint security enhancements offer tailored solutions to address these cybersecurity challenges in emerging markets (Dacre & Kockum, 2022a; Tariq et al., 2024).

#### 2.2 Fluctuations in Interest Rates

Fluctuations in interest rates represent a considerable challenge for commercial banks in managing credit risk (Kithinji, 2010; Santomero, 1997). Shifts in interest rates influence not only borrowers' ability to repay loans but also the value of the collateral securing those loans (Cerqueiro et al., 2016). In low interest rate environments, borrowers may be inclined to take on excessive debt, increasing their likelihood of default (Krishnamurthy, 2010). Conversely, rising interest rates can strain borrowers' financial capacity, resulting in higher NPLs and greater credit losses for banks (Louzis et al., 2012; Serkbayeva et al., 2024; Siddique et al., 2022).

A considerable body of research has examined the interplay between interest rates and credit risk across different financial systems (Crouhy et al., 2000; Naili & Lahrichi, 2022; Thakor, 1996). For example, studies from developed markets, including Europe, the United States, and Canada, have found that low short-term interest rates soften lending standards for both household and corporate loans (Maddaloni & Peydró, 2011; Paligorova & Sierra, 2012). This effect is particularly pronounced in mortgage lending and is amplified by securitisation activities and weak bank capital supervision (Maddaloni &

Peydró, 2011). Banks tend to decrease credit standards and increase exposure to high-yield, high-risk loans in search of better returns (Gambacorta, 2009). This risk-taking channel of monetary policy may exacerbate the effects of traditional transmission mechanisms, potentially leading to excessive credit creation (Paligorova & Sierra, 2012). Notably, countries with softer lending standards before the 2007-09 financial crisis, associated with negative Taylor rule residuals, experienced worse economic performance in its aftermath (Maddaloni & Peydró, 2011).

In emerging markets, the challenges associated with interest rate fluctuations are often magnified. These economies are particularly susceptible to currency volatility, inflationary pressures, and limited access to effective hedging mechanisms, all of which compound the risks posed by interest rate movements (Saxena & Villar, 2008). Such vulnerabilities frequently lead firms to borrow in foreign currencies, exposing them to currency risk premia and heightening their sensitivity to global monetary shifts (Chinoy, 2002). The interest rate sensitivity of emerging market debt is also highly dynamic, exhibiting regimedependent behaviour that shifts from positive to negative during times of crisis (Gubareva & Borges, 2022). This is shaped by factors such as the business cycle, debt maturity, and the creditworthiness of borrowers (Gubareva & Borges, 2016a, 2022). Small and mediumsized enterprises which are the pillar of many emerging market economies, are particularly vulnerable to these fluctuations, as they often lack the financial resilience and access to risk mitigation tools afforded by larger firms (Hadiat & Anjani). As such, emerging markets face both idiosyncratic and systemic risks, with external shocks such as US monetary policy narrowing triggering correlated declines in local currencies and escalating financial pressures (Wise, 2017). Additionally, real interest rate volatility exerts significant effects on key economic variables, including output, consumption, investment, and labour productivity, expanding the challenges faced by small open economies (Fernández-Villaverde et al., 2011). Evidence from Turkey further underscores these risks, in that heightened interest rate volatility has a pronounced negative impact on bank profitability, reinforcing the critical need for robust interest rate risk management frameworks (Kasman & Kasman, 2016).

#### 2.3 Inadequate Resources and Expertise

The role of human capital in credit risk management has been consistently emphasised in academic research (Bauer et al., 2009; Dacre et al., 2015; Dacre & Kockum, 2022b;

Gkogkidis & Dacre, 2020a; John et al., 2015; Stamolampros & Symitsi, 2022). Studies indicate that banks with experienced and proficient risk managers are better equipped to navigate credit risk challenges, particularly during periods of financial instability (Aebi et al., 2012; Hazaa et al., 2021). For instance, research examining bank performance during the global financial crisis revealed that institutions with stronger risk management expertise demonstrated greater resilience (Aebi et al., 2012; Dacre, 2024c; Mateev, Usman Tariq, et al., 2023).

In addition to human expertise, technological resources and data infrastructure play a critical role in effective credit risk management (Dacre, AlJaloudi, Thiam, et al., 2024; Ng & Kwok, 2017; Nobanee et al., 2024). Advanced tools such as artificial intelligence, machine learning algorithms, big data analytics, and predictive modelling can enhance banks' ability to assess creditworthiness and monitor loan performance (Al-Mhdawi, Qazi, et al., 2023; Antonopoulou & Dacre, 2021; Bhatore et al., 2020; Corbin et al., 2024; Dacre, 2024a; Dacre et al., 2020; Hsu et al., 2021a). These technologies provide granular insights into borrower behaviour, enabling more accurate risk assessments and proactive mitigation strategies. However, the implementation of such tools relies heavily on the availability of reliable technological infrastructure and high-quality data (Brookes et al., 2020; Kockum & Dacre, 2021; Ogunmokun et al., 2023).

Emerging economies frequently encounter barriers to adopting these nascent technologies, including limited access to reliable data, emerging technological ecosystems, and financial constraints that incumber investment in modern risk management systems (Kandasamy et al., 2023). For example, a case study examining credit risk management in Nepalese banks highlights the significant impact of resource and expertise gaps on risk management practices (Poudel, 2012). The study identified critical challenges, including a shortage of skilled human resources and inadequate technological capabilities.

In contextualising these challenges, resource scarcity appears to be a recurrent concern in many emerging markets, where institutions may face difficulties in securing sufficient funding for staff development and the adoption of advanced risk management technologies (Hoskisson et al., 2000). This fragmentation of resources suggests the possibility of a classic transition, wherein banks could benefit from enhancing internal capacity while also exploring opportunities to leverage external support mechanisms.

#### 2.4 Changes in Regulatory Frameworks

Shifts in regulatory frameworks introduce both opportunities and challenges for commercial banks in managing credit risk (Van Greuning & Bratanovic, 2020). The introduction of new regulations or amendments to existing ones often requires banks to reassess their risk management practices, operational processes, and reporting systems (Barakat & Hussainey, 2013; Dacre, Eggleton, Gkogkidis, et al., 2021). While such changes are often designed to enhance financial stability and resilience, their implementation is not without cost. Non-compliance risks include financial penalties, reputational harm, and potential disruptions to banking operations (Allen & Wood, 2006).

The impact of regulatory changes on credit risk management has been a subject of interest in the literature (Jackson & Perraudin, 2000). For example, while the Basel III regulatory framework has been lauded for strengthening global banking systems through enhanced capital requirements and liquidity standards, critics argue that its application may disproportionately burden smaller banks and those in emerging markets (Bo, 2024; Duho et al., 2023; Frait & TomŠÍk, 2014). Such measures, while theoretically beneficial, may inadvertently incentivise risk-taking in certain contexts. Banks under regulatory pressure to meet stringent capital thresholds might seek high-yield, high-risk investments to sustain profitability, potentially undermining the very stability these regulations aim to ensure (Berger & Demirgüç-Kunt, 2021; Calem & Rob, 1999).

In emerging markets, the challenge of regulatory compliance may reflect broader systemic vulnerabilities rather than isolated policy issues (Abdel-Baki, 2012; Claessens, 2010). For instance, while research from sub-Saharan Africa suggests that regulatory reforms have supported bank stability in some cases, their impact has been uneven and, at times, counterproductive (Gottschalk, 2014). Frequent policy changes or poorly calibrated reforms can create uncertainty, discouraging long-term planning and innovation among banks (Kamasa et al., 2023). Additionally, the administrative burden of compliance may divert resources away from core banking functions, such as lending, particularly in economies with constrained financial infrastructure (Asiama & Amoah, 2019).

# 2.5 Lack of System Integration

Efficient coordination and seamless data flow between departments, such as credit, risk, and finance, are considered essential for effective credit risk management. Studies suggest

that the lack of integration between these departments may lead to inconsistencies in data, fragmented communication channels, and delays in decision-making processes, which can increase exposure to credit risk (Alrabiah, 2018; Yanenkova et al., 2021). Research highlights the role of system integration in enabling timely and accurate risk assessments, emphasising the potential benefits of cohesive data management frameworks within banks (Moges et al., 2012; Pontin & Dacre, 2024; Simiyu, 2008).

Empirical findings underscore the significance of integrated systems in enhancing credit risk management practices. Studies have demonstrated how IT integration can improve a bank's capacity to aggregate and analyse data effectively, enabling more precise risk assessments and timely interventions (Alrabiah, 2018; Dacre, Yan, Dong, et al., 2024; Gubareva & Borges, 2016b; Hsu et al., 2021b; Xu et al., 2023). For example, research focusing on emerging markets has revealed that IT integration enhances coordination between credit and risk departments, promoting more consistent monitoring of credit exposures (Addy et al., 2024; Wagdi & Tarek, 2022). Integrated systems also support the automation of routine processes, minimising errors and inefficiencies that can heighten credit risk (Dacre et al., 2014; Dacre et al., 2022; Nahar et al., 2024; Nikolaidou & Vogiazas, 2017).

Despite these potential benefits, research also highlights the challenges associated with achieving system integration, particularly in resource-constrained environments. In emerging markets, fragmented legacy systems, data silos, and inadequate IT infrastructure are often cited as obstacles to digital transformation (Dacre et al., 2019; Dong & Dacre, 2024; Jameaba, 2024; Manh et al., 2024). Studies on emerging market banks have documented how these limitations can hinder the development of cohesive digital ecosystems, which are crucial for seamless inter-departmental coordination (Naimi-Sadigh et al., 2022). Moreover, cultural and organisational factors, such as resistance to change and a lack of technical expertise, can complicate efforts to implement integrated systems (Campino et al., 2021; Ofosu-Ampong, 2021).

#### 2.6 Summary of Key Studies

The literature reviewed highlights the critical role of credit risk management in influencing financial stability and bank performance across various contexts. Studies in developed markets have established clear relationships between factors such as Non-Performing Loan

(NPL) ratios, macroeconomic conditions, and profitability. However, research focusing on emerging markets reveals unique challenges, including heightened exposure to external shocks, emergent financial systems, and evolving regulatory frameworks (Table 3).

Table 3: Key Contributions to Credit Risk Literature

Author(s) and Year	Key Findings	Geographical Focus
Al-Tamimi & Al-Mazrooei	Jordanian banks limits in risk	Jordan
(2007); AlJaloudi et al. (2024);	management, with credit risk affecting	
Alshatti (2015)	profitability.	
Anderson (2007); Ghosh	Fintech adoption (i.e., advanced	Emerging
(2015); Serkbayeva et al.	analytics, risk assessment tools)	Markets
(2024)	enhances credit scoring accuracy in	
	emerging markets.	
Awijen et al. (2023); Ghosh	Macroeconomic factors (i.e., GDP	Developed
(2015); Mpofu & Nikolaidou	growth, inflation, unemployment) are	Markets
(2018); Reinhart & Rogoff	critical determinants of credit risk	
(2011)	exposure.	
Bayar (2019); Berger &	Higher Non-Performing Loan (NPL)	Developed
DeYoung (1997); Chen et al.	ratios reduce profitability and weaken	Markets
(2021); Louzis et al. (2012)	financial performance.	
Das & Ghosh (2007); Manda	Bank size, loan growth, and	India, Pakistan
& Rani (2019); Siddique et al.	macroeconomic conditions are	
(2022)	significant contributors to credit risk.	
Hawkins & Mihaljek (2001);	Emerging markets face unique	Emerging
Bekaert & Harvey (2003)	challenges such as weak legal	Markets
	frameworks and underdeveloped	
	financial systems.	
Kamiya et al. (2021);	Cybersecurity breaches compromise	Global
Mohamed et al. (2024); Novak	credit risk management through data	
& Vilceanu (2019)	loss and operational disruptions.	
Karim et al. (2010)	Effective credit risk management	Malaysia
	mitigates financial performance risks.	
Kasman & Kasman (2016);	Interest rate volatility negatively impacts	Emerging
Gubareva & Borges (2022)	bank profitability, with heightened	Markets
,	sensitivity in emerging markets.	
Moussa (2015); Siddique et al.	Macroeconomic volatility and lack of	South Asia,
(2022)	credit information amplify default risks	North Africa

The literature underscores the importance for Jordanian banks to refine and adapt their credit risk management strategies to align with the country's unique economic, institutional, and technological realities. The challenges identified, ranging from evolving risk management practices to heightened vulnerability to external shocks, highlight the importance of strengthening internal controls, ensuring regulatory compliance, and advancing technological integration. Additionally, the specific limitations faced by

Jordanian banks, such as resource constraints, developing credit information systems, and the gradual adoption of fintech solutions, mandate tailored strategies that not only mitigate credit risk but also enhance overall bank performance within this emerging market context.

# 3. Methodology

Our study employs an ongoing methodological framework to investigate the credit risk challenges faced by Jordanian banks and their implications for financial outcomes. The research design integrates exploratory and descriptive elements in order to provide an iterative analysis of the phenomenon(Creswell & Clark, 2017). We combine a systematic review of existing literature with empirical data collection methods, ensuring a triangulated and adaptive approach that evolves as the research progresses. The literature review forms the theoretical foundation, identifying key credit risk challenges and contextualising them within the unique dynamics of emerging markets (Booth et al., 2021). This informs the design of our ongoing data collection strategies, which are continually refined to address emerging insights and gaps.

The empirical phase of our research began with the deployment of a structured questionnaire to banking professionals in Jordan. The questionnaire was designed to gather detailed data on the prevalence of key credit risk challenges, such as cybersecurity threats, interest rate fluctuations, resource constraints, regulatory changes, and system integration issues (Dillman et al., 2014). It also aims to capture insights into the current risk management practices employed by banks and the barriers they face in addressing these challenges. The questionnaire development process followed established practices, ensuring clarity, relevance, and reliability in the design of survey instruments (DeVellis & Thorpe, 2021; Reynolds & Dacre, 2019). The target population for the questionnaire comprises senior executives, risk managers, and credit officers from Jordanian commercial banks. A purposive sampling technique was employed to select participants with substantial knowledge and experience in credit risk management (Etikan et al., 2016). The questionnaire was distributed electronically to the selected participants, and follow-up reminders were sent to ensure a satisfactory response rate. Data collection through the questionnaire is ongoing, and responses are analysed to identify emerging trends and areas that require further investigation.

# 4. Results and Discussion

The preliminary analysis of the data collected reveals several key credit risk challenges faced by commercial banks in Jordan and their potential implications for banks' performance. These challenges can be broadly categorised into five main themes: cybersecurity threats, fluctuations in interest rates, inadequate resources and expertise, changes in regulatory frameworks, and the lack of system integration within banks' internal departments.

Cybersecurity threats emerge as a significant concern for Jordanian banks, mirroring the global trend highlighted in the literature (Kamiya et al., 2021; Zou et al., 2018). The increasing digitalisation of financial services has exposed banks to new types of risks, such as data breaches and cyberattacks. Respondents emphasise the need for continuous investment in effective security measures and vigilant monitoring of potential vulnerabilities. This finding underscores the importance of proactive cybersecurity risk management in the Jordanian banking sector, as the literature suggests that cybersecurity incidents can have severe consequences for banks' financial performance and reputation (Eling et al., 2021; Grima et al., 2023). Moreover, the legacy infrastructure in many Jordanian banks may exacerbate their vulnerability, highlighting the critical need for upgrading technological defences (Ani et al., 2017).

Fluctuations in interest rates are identified as another critical challenge for Jordanian banks in managing credit risk. This finding is consistent with the literature, which highlights the impact of interest rate changes on borrowers' ability to repay loans and the value of collateral (Kasman & Kasman, 2016; Louzis et al., 2012). Respondents stress the need for banks to develop effective strategies to mitigate the impact of interest rate fluctuations on their loan portfolios and overall performance. Given the potential amplification of interest rate risk in emerging markets due to factors such as currency volatility and inflationary pressures, this finding emphasises the importance of effective interest rate risk management practices in the Jordanian banking sector (Gubareva & Borges, 2022; Louzis et al., 2012).

Inadequate resources and expertise in credit risk management are identified as significant challenges by the respondents, echoing the concerns raised in the literature (Dacre, AlJaloudi, Al-Mhdawi, et al., 2024; Poudel, 2012). The lack of skilled personnel and

advanced risk management tools can hinder banks' ability to effectively assess and manage credit risk. Respondents highlight the need for investing in capacity building and technology upgrades to enhance credit risk management capabilities (Dacre et al., 2018; Gkogkidis & Dacre, 2020b, 2021, 2023). This finding suggests that addressing the resource and expertise gaps is crucial for Jordanian banks to strengthen their credit risk management practices and improve their performance (Bhatore et al., 2020; Mateev, Sahyouni, et al., 2023). As prior research highlights, resource constraints are a recurring issue in many emerging markets, where financial institutions often struggle to balance immediate operational demands with long-term investments in expertise and technology (Ng & Kwok, 2017).

Changes in regulatory frameworks, such as the introduction of new Basel standards or local regulations, are also identified as challenges for Jordanian banks. This finding is in line with the literature, which emphasises the impact of regulatory changes on banks' risk management practices, processes, and reporting systems (Abdel-Baki, 2012; Van Greuning & Bratanovic, 2020). Respondents stress the importance of proactive engagement with regulators and staying informed about upcoming regulatory changes. This finding highlights the need for Jordanian banks to adapt to the evolving regulatory landscape and ensure compliance to maintain financial stability (Frait & TomŠík, 2014). Moreover, frequent regulatory adjustments can introduce uncertainty and operational burdens, which are particularly challenging for smaller banks with limited resources (Gottschalk, 2014; Jackson & Perraudin, 2000).

Finally, the lack of system integration within banks' internal departments is identified as a challenge hindering effective credit risk management. This finding is consistent with the literature, which emphasises the importance of seamless data flow and coordination between various departments for timely and accurate risk assessment and decision-making (Alrabiah, 2018; Yanenkova et al., 2021). Respondents underscore the need for improving system integration to enhance credit risk management practices. This suggests that addressing the challenges related to system integration is crucial for Jordanian banks to optimise their credit risk management processes and improve their overall performance (Addy et al., 2024; Gubareva & Borges, 2016b). Enhanced IT integration has been shown to reduce operational inefficiencies and improve risk assessment accuracy, underscoring its importance in managing credit risk effectively in emerging markets (Nahar et al., 2024).

# 5. Conclusion

This paper has sought to examine the relationship between credit risk challenges and banks' performance in emerging markets, focusing on the case of Jordan. The preliminary findings, based on a review of the literature and an analysis of questionnaires, highlight several key credit risk challenges faced by Jordanian commercial banks, including cybersecurity threats, fluctuations in interest rates, inadequate resources and expertise, regulatory changes, and a lack of system integration (Table 4).

Table 4: Credit Risk Challenges and Mitigation Strategies

Key Credit Risk Challenge	Description	Implications for Banks	Potential Strategies for Mitigation
Cybersecurity Threats	Increased risk of data breaches and cyberattacks due to digitalisation of financial services.	Financial losses, reputational damage, erosion of customer trust, and operational disruptions.	Investment in advanced cybersecurity measures, real-time threat detection systems, multi-factor authentication, and staff training.
Fluctuations in Interest Rates	Changes in interest rates affecting loan repayment capacity and collateral value.	Higher default rates, reduced profitability, and instability in loan portfolios.	Implementation of dynamic risk management frameworks, use of stress testing and scenario analysis, and development of robust hedging mechanisms.
Inadequate Resources and Expertise	Lack of skilled personnel and advanced tools for credit risk management.	Reduced efficiency in risk assessment, increased exposure to credit risk, and lower performance.	Capacity-building programmes, adoption of advanced analytics and machine learning tools, and investment in human capital development.
Regulatory Changes	Introduction of new regulations or amendments, such as Basel standards.	Increased compliance costs, operational strain, and uncertainty for smaller institutions.	Proactive engagement with regulators, early adoption of compliance measures, and collaborative policy dialogue to align resources.
Lack of System Integration	Fragmented systems and poor coordination between internal departments, such as risk and finance.	Inefficiencies in data flow, delays in decision-making, and inconsistent risk assessments.	Integration of IT systems, adoption of collaborative workflows, and implementation of cohesive data management frameworks.

The study contributes to the existing body of knowledge by providing insights into the specific credit risk challenges encountered by banks within an emerging market context. The alignment of these identified challenges with themes discussed in the literature

underscores the relevance and significance of these issues in the broader domain of credit risk management. Therefore, by focusing on the Jordanian banking sector, this research offers a deeper understanding of the unique challenges faced by banks operating in a specific emerging market.

Moreover, the findings of this study have significant implications for practice and policymaking. The identified credit risk challenges can have profound consequences for banks' performance, as they may impede effective credit risk management and expose banks to potential losses. Addressing these challenges mandates a proactive and comprehensive approach, involving investments in technology, capacity building, and effective risk management practices. Banks should therefore prioritise investments in cybersecurity measures, interest rate risk management strategies, and capacity building programmes to enhance the skills and expertise of their risk management personnel. Additionally, banks should focus on improving system integration to optimise data flow and coordination between internal departments, thereby enhancing the efficiency and effectiveness of credit risk management processes. Policymakers should consider implementing policies and regulations that promote financial stability, encourage innovation, and foster the development of a skilled workforce (Dacre, 2024b; Dacre, Eggleton, Cantone, et al., 2021; Tite et al., 2021b). Regulators should engage in continuous dialogue with banks to ensure that regulatory requirements are well understood and that banks have the necessary resources and capabilities to comply with these requirements.

However, it is important to acknowledge the limitations of this study. The findings are based on a preliminary analysis and may not fully capture the complexity of the credit risk management landscape in Jordan. Further research is required to validate and expand upon these initial insights, incorporating a larger sample of banks and employing more advanced analytical techniques. Future research could also explore the interplay between credit risk challenges and other factors, such as corporate governance, macroeconomic conditions, and market structure, to gain a more comprehensive understanding of the determinants of banks' performance in emerging markets.

In conclusion, while our findings align with broader trends observed globally, they reveal a distinct intersection of systemic vulnerabilities and emerging risks unique to the Jordanian banking sector. For instance, the convergence of resource constraints, regulatory

evolution, and technological lag creates a compounded effect, where challenges amplify each other rather than existing in isolation. Cybersecurity vulnerabilities are not just a technological issue but are exacerbated by inadequate expertise and resource deficits, while regulatory pressures are more acutely felt in institutions already struggling with system fragmentation. This dynamic interplay of challenges underscores the critical impetus for a holistic approach to credit risk management, where solutions are integrated across dimensions rather than addressing individual issues in silos. Furthermore, the findings suggest that the Jordanian banking sector, while sharing some commonalities with other emerging markets, operates within a context shaped by unique economic, institutional, and technological factors.

#### References

- Abdel-Baki, M. A. (2012). The impact of Basel III on emerging economies. *Global Economy Journal*, 12(2), 1850256.
- Abdou, D. M. S., & Alarabi, Y. (2024). The dynamics behind private banking growth in Egypt. *Future Business Journal*, 10(1), 1.
- Abidin, M. A. Z., Nawawi, A., & Salin, A. S. A. P. (2019). Customer data security and theft: a Malaysian organization's experience. *Information & Computer Security*, 27(1), 81-100.
- Addy, W. A., Ugochukwu, C. E., Oyewole, A. T., Ofodile, O. C., Adeoye, O. B., & Okoye, C. C. (2024). Predictive analytics in credit risk management for banks: A comprehensive review. *GSC Advanced Research and Reviews*, 18(2), 434-449.
- Aebi, V., Sabato, G., & Schmid, M. (2012). Risk management, corporate governance, and bank performance in the financial crisis. *Journal of Banking & Finance*, 36(12), 3213-3226.
- Al-Mhdawi, M. K. S., Dacre, N., Brito, M., Baxter, D., Xu, K., & Young, C. (2023). An Agile compliance framework for the European Cooperation for Space Standardization. *IEEE Aerospace*, 12. https://doi.org/10.1109/AERO55745.2023.10115917
- Al-Mhdawi, M. K. S., O'Connor, A., Qazi, A., Dacre, N., & Al-Saedi, M. W. (2023). A Proposed Fuzzy-based Optimisation Model for Evaluating Construction Projects' Risk Response Strategies. *Applications of Statistics and Probability in Civil Engineering*, 9. https://www.tara.tcd.ie/handle/2262/103651
- Al-Mhdawi, M. K. S., O'connor, A., Qazi, A., Rahimian, F., & Dacre, N. (2024). Review of studies on risk factors in critical infrastructure projects from 2011 to 2023. *Smart and Sustainable Built Environment*, 35. https://doi.org/10.1108/SASBE-09-2023-0285
- Al-Mhdawi, M. K. S., Qazi, A., Alzarrad, A., Dacre, N., Rahimian, F., Buniya, M. K., & Zhang, H. (2023). Expert Evaluation of ChatGPT Performance for Risk

- Management Process Based on ISO 31000 Standard. *SSRN Electronic Journal*, 6. https://doi.org/10.2139/ssrn.4504409
- Al-Smadi, M. O., & Al-Wabel, S. A. (2011). The impact of e-banking on the performance of Jordanian banks. *Journal of internet banking and commerce*, 16(2), 1.
- Al-Tamimi, H. A. H., & Al-Mazrooei, F. M. (2007). Banks' risk management: a comparison study of UAE national and foreign banks. *The journal of risk finance*, 8(4), 394-409.
- Aldasoro, I., Gambacorta, L., Giudici, P., & Leach, T. (2020). Operational and cyber risks in the financial sector.
- Aldasoro, I., Gambacorta, L., Giudici, P., & Leach, T. (2022). The drivers of cyber risk. *Journal of Financial Stability*, 60, 100989.
- AlJaloudi, O., Thiam, M., Abdel Qader, M., Al-Mhdawi, M. K. S., Qazi, A., & Dacre, N. (2024). Examining the Integration of Generative AI Models for Improved Risk Management Practices in the Financial Sector *International Finance*, 14.
- Allen, W. A., & Wood, G. (2006). Defining and achieving financial stability. *Journal of Financial Stability*, 2(2), 152-172.
- Alrabiah, A. (2018). Optimal regulation of banking system's advanced credit risk management by unified computational representation of business processes across the entire banking system. *Cogent Economics & Finance*, 6(1), 1486685.
- Alshatti, A. S. (2015). The effect of credit risk management on financial performance of the Jordanian commercial banks. *Investment management and financial innovations*(12,№ 1 (contin. 2)), 338-345.
- Alzuod, M., & Alqhaiwi, L. A. (2022). The role of strategic flexibility in enhancing crisis management in the commercial banking sector during the COVID-19 pandemic. *Corporate Governance and Organizational Behavior Review*, 6(2).
- Analytica, O. (2022). Optus breach spurs Australian cybersecurity reform. *Emerald Expert Briefings* (oxan-db).
- Anderson, R. (2007). The credit scoring toolkit: theory and practice for retail credit risk management and decision automation. Oxford university press.
- Ani, U. P. D., He, H., & Tiwari, A. (2017). Review of cybersecurity issues in industrial critical infrastructure: manufacturing in perspective. *Journal of Cyber Security Technology*, *I*(1), 32-74.
- Antonopoulou, K., & Dacre, N. (2021). Exploring Diffusion Characteristics that Influence Serious Games Adoption Decisions. *Innovation in Information Infrastructures*, 10. https://doi.org/10.48550/arXiv.2105.01745
- Arcuri, M. C., Brogi, M., & Gandolfi, G. (2017). Cyber risk: A big challenge in developed and emerging markets. In *Identity theft: breakthroughs in research and practice* (pp. 292-307). IGI Global.
- Asiama, R. K., & Amoah, A. (2019). Non-performing loans and monetary policy dynamics in Ghana. *African Journal of Economic and Management Studies*, 10(2), 169-184.
- Audi, M., & Al-Masri, R. (2024). Examining the Impacts of Regulatory Framework on Risk in Commercial Banks in Emerging Economies.

- Avery, R. B., Calem, P. S., & Canner, G. B. (2004). Credit report accuracy and access to credit. *Fed. Res. Bull.*, *90*, 297.
- Awijen, H., Ben Zaied, Y., & Hunjra, A. I. (2023). Systematic and Unsystematic Determinants of Sectoral Risk Default Interconnectedness. *Computational Economics*, 62(2), 561-587.
- Ayayi, A. G. (2012). Credit risk assessment in the microfinance industry: An application to a selected group of Vietnamese microfinance institutions and an extension to East Asian and Pacific microfinance institutions 1. *Economics of Transition*, 20(1), 37-72.
- Barakat, A., & Hussainey, K. (2013). Bank governance, regulation, supervision, and risk reporting: Evidence from operational risk disclosures in European banks. *International Review of Financial Analysis*, 30, 254-273.
- Barber, C., Dacre, N., & Dong, H. (2021). Reframing Project Management Process Paralysis: An Autoethnographic Study of the UK Fire Service. *Advanced Project Management*, 21(6), 4. https://doi.org/10.31235/osf.io/hxm68
- Bauer, R., Derwall, J., & Hann, D. (2009). Employee relations and credit risk. *Available at SSRN 1483112*.
- Baxter, D., Dacre, N., Dong, H., & Ceylan, S. (2023). Institutional challenges in agile adoption: Evidence from a public sector IT project. *Government Information Quarterly*, 40(4), 15. https://doi.org/10.1016/j.giq.2023.101858
- Bayar, Y. (2019). Macroeconomic, institutional and bank-specific determinants of non-performing loans in emerging market economies: A dynamic panel regression analysis. *Journal of Central Banking Theory and Practice*, 8(3), 95-110.
- Beerman, J., Berent, D., Falter, Z., & Bhunia, S. (2023). A review of colonial pipeline ransomware attack. 2023 IEEE/ACM 23rd International Symposium on Cluster, Cloud and Internet Computing Workshops (CCGridW),
- Bekaert, G., & Harvey, C. R. (2003). Emerging markets finance. *Journal of empirical finance*, 10(1-2), 3-55.
- Berger, A. N., & Demirgüç-Kunt, A. (2021). Banking research in the time of COVID-19. *Journal of Financial Stability*, 57, 100939.
- Berger, A. N., & DeYoung, R. (1997). Problem loans and cost efficiency in commercial banks. *Journal of Banking and Finance*, 21(6), 849-870. https://doi.org/10.1016/S0378-4266(97)00003-4
- Beshouri, C. P., & Gravråk, J. (2010). Capturing the promise of mobile banking in emerging markets. *McKinsey Quarterly*, 3(8).
- Bhatore, S., Mohan, L., & Reddy, Y. R. (2020). Machine learning techniques for credit risk evaluation: a systematic literature review. *Journal of Banking and Financial Technology*, 4(1), 111-138.
- Bi, S., & Bao, W. (2024). Innovative Application of Artificial Intelligence Technology in Bank Credit Risk Management. *International Journal of Global Economics and Management*, 2(3), 76-81.
- Biddle, N., Gray, M., & McEachern, S. (2022). Public exposure and responses to data breaches in Australia: October 2022.

- Bo, W. (2024). The Impact of Macroeconomic Factors on Credit Risk Management in Developing Country Banks: An Analysis Based on Basel III. *Financial Engineering and Risk Management*, 7(5), 97-105.
- Booth, A., James, M.-S., Clowes, M., & Sutton, A. (2021). Systematic approaches to a successful literature review.
- Brookes, N., Lattuf Flores, L., Dyer, R., Stewart, I., Wang, K., & Dacre, N. (2020). *Project Data Analytics: The State of the Art and Science*. Association for Project Management. https://doi.org/10.61175/wgsv1658
- Bukth, T., & Huda, S. S. (2017). *The soft threat: The story of the Bangladesh bank reserve heist*. SAGE Publications: SAGE Business Cases Originals.
- Calem, P., & Rob, R. (1999). The impact of capital-based regulation on bank risk-taking. *Journal of Financial Intermediation*, 8(4), 317-352.
- Campino, J., Brochado, A., & Rosa, Á. (2021). Digital business transformation in the banking sector. In *Research Anthology on Concepts, Applications, and Challenges of FinTech* (pp. 186-215). IGI Global.
- Cele, N. N., & Kwenda, S. (2024). Do cybersecurity threats and risks have an impact on the adoption of digital banking? A systematic literature review. *Journal of Financial Crime*.
- Cerqueiro, G., Ongena, S., & Roszbach, K. (2016). Collateralization, bank loan rates, and monitoring. *The Journal of Finance*, 71(3), 1295-1322.
- Chari, A. (2023). Global risk, non-bank financial intermediation, and emerging market vulnerabilities. *Annual Review of Economics*, 15(1), 549-572.
- Chen, W.-D., Chen, Y., & Huang, S.-C. (2021). Liquidity risk and bank performance during financial crises. *Journal of Financial Stability*, *56*, 100906.
- Chinoy, S. Z. (2002). Currency risk premia and unhedged, foreign-currency borrowing in emerging markets. Stanford University.
- Claessens, S. (2010). The financial crisis: Policy challenges for emerging markets and developing countries. *Margin: The Journal of Applied Economic Research*, 4(2), 177-196.
- Claessens, S., Cong, L. W., Moshirian, F., & Park, C.-Y. (2024). Opportunities and challenges associated with the development of FinTech and Central Bank Digital Currency. In (pp. 101280): Elsevier.
- Corbin, D., Marqui, A., & Dacre, N. (2024). The Intersection of Artificial Intelligence and Project Management in UK Construction: An Exploration of Emerging Trends, Enablers, and Barriers. *European Operations Management Association*, 16. https://doi.org/10.2139/ssrn.5015982
- Costa, O., Khan, J., Levy, C., Natale, A., & Tanrikulu, O. (2014). Risk in emerging markets. *USA: McKinsey & Company*, 19.
- Creswell, J. W., & Clark, V. L. P. (2017). Designing and conducting mixed methods research. Sage publications.
- Crouhy, M., Galai, D., & Mark, R. (2000). A comparative analysis of current credit risk models. *Journal of Banking & Finance*, 24(1-2), 59-117.

- Dacre, N. (2024a). Cadres Conceptuels et Technologiques de la Pensée Systémique et de l'Intelligence Artificielle: Vers une Gouvernance Adaptative en Gestion de Projets. *Hyper Articles en Ligne*(hal-04775927v1), 17. https://hal.science/hal-04775927
- Dacre, N. (2024b). *Diversity in project teams: purpose, progress and delivery*. Association for Project Management. https://www.apm.org.uk/v2/media/yskhsac5/diversity-in-project-teams-purpose-progress-and-delivery.pdf
- Dacre, N. (2024c). L'Innovation de l'Apprentissage en Ligne dans l'Éducation Post-Pandémique: Gamification et Jeu Sérieux pour le Développement des Compétences en Gestion de Projet. *Hyper Articles en Ligne*(hal-04754963), 16. https://hal.science/hal-04754963v1
- Dacre, N., AlJaloudi, O., Al-Mhdawi, M. K. S., Thiam, M., Abdel Qader, M., & Qazi, A. (2024). Examining the Impact of Credit Risk Management Challenges on Bank Performance: The Mediating Role of Best Practices. *Credit Risk*, 21.
- Dacre, N., AlJaloudi, O., Thiam, M., Abdel Qader, M., Al-Mhdawi, M. K. S., & Qazi, A. (2024). Capturing Fintech's Role in Optimising Financial Risk Management. *Operational Research Society*, 9. https://doi.org/10.2139/ssrn.4985879
- Dacre, N., Constantinides, P., & Nandhakumar, J. (2014). Instantiation of Organisational Routines in Cross-Expertise Collaborative Enterprise Systems. *Process Organization Studies*, 5. https://doi.org/10.31235/osf.io/ky8s7
- Dacre, N., Constantinides, P., & Nandhakumar, J. (2015). How to Motivate and Engage Generation Clash of Clans at Work? Emergent Properties of Business Gamification Elements in the Digital Economy. *International Gamification for Business*, 11. https://doi.org/10.48550/arXiv.2103.12832
- Dacre, N., Dong, H., Gkogkidis, V., & Kockum, F. (2022). Innovative Strategies for Distance Learning: Gamification, Serious Play, and Miro in the Development of Project Management Competencies *Operational Research Society*, 22. https://doi.org/10.2139/ssrn.4997288
- Dacre, N., Eggleton, D., Cantone, B., & Gkogkidis, V. (2021). Why People Skills Lead to Project Success: Towards Dynamic Conditions for People Skills and Leadership in Project Management *Project*, 307, 14. https://doi.org/10.2139/ssrn.4998962
- Dacre, N., Eggleton, D., Gkogkidis, V., & Cantone, B. (2021). Expanding the Paradigm of Project Success: A Review of Diversity as a Critical Success Condition in Project Management *SSRN Electronic Journal*, 23. https://doi.org/10.2139/ssrn.5001594
- Dacre, N., Gkogkidis, V., & Jenkins, P. (2018). Co-Creation of Innovative Gamification Based Learning: A Case of Synchronous Partnership. *Society for Research into Higher Education*, 6. https://doi.org/10.48550/arXiv.2103.13273
- Dacre, N., & Kockum, F. (2022a). Artificial Intelligence in Project Management: A review of AI's usefulness and future considerations for the project profession. Association for Project Management. https://doi.org/10.61175/DOGX9829
- Dacre, N., & Kockum, F. (2022b). Strategic Project Leadership: The Case for Integrating Systems Thinking and Artificial Intelligence *Operational Research Society*, 20. https://doi.org/10.2139/ssrn.5015629

- Dacre, N., Kockum, F., & Senyo, P. (2020). Transient Information Adaptation of Artificial Intelligence: Towards Sustainable Data Processes in Complex Projects. *British Academy of Management*, 2(63), 9. https://doi.org/10.48550/arXiv.2104.04067
- Dacre, N., Senyo, P., & Reynolds, D. (2019). Is an Engineering Project Management Degree Worth it? Developing Agile Digital Skills for Future Practice. *Engineering Education Research*, 8. https://doi.org/10.31235/osf.io/4b2gs
- Dacre, N., Yan, J., Dong, H., Al-Mhdawi, M. K. S., & Frei, R. (2024). Assessing the Influence of Industry 5.0 on Supply Chain Integration and Performance: An Empirical Analysis within the Manufacturing Sector. *Operational Research Society*, 12. https://doi.org/10.36227/techrxiv.173091022.25517688/v1
- Dacre, N., Yan, J., Frei, R., Al-Mhdawi, M. K. S., & Dong, H. (2024). Advancing sustainable manufacturing: a systematic exploration of industry 5.0 supply chains for sustainability, human-centricity, and resilience. *Production Planning & Control*, 31. https://doi.org/10.1080/09537287.2024.2380361
- Das, A., & Ghosh, S. (2007). Determinants of credit risk in Indian state-owned banks: An empirical investigation.
- De Marco, F. (2019). Bank lending and the European sovereign debt crisis. *Journal of Financial and Quantitative Analysis*, 54(1), 155-182.
- DeVellis, R. F., & Thorpe, C. T. (2021). *Scale development: Theory and applications*. Sage publications.
- Dillman, D. A., Smyth, J. D., & Christian, L. M. (2014). *Internet, phone, mail, and mixed-mode surveys: The tailored design method.* John Wiley & Sons.
- Dong, H., & Dacre, N. (2024). Commutable Strategies and Institutional Dynamics in Agricultural Co-operatives: A Project Management Perspective on Crisis Resilience and Sustainability *Earth and Space Science*, 22. https://doi.org/10.22541/essoar.173134645.56504196/v1
- Dong, H., Dacre, N., & Bailey, A. (2021a). Sustainability in Complex Agriculture Projects:

  A Study of Agile Agricultural Co-operative Institutions. *British Academy of Management*, 10. https://doi.org/10.2139/ssrn.3879454
- Dong, H., Dacre, N., & Bailey, A. (2021b). Sustainable Agile Project Management in Complex Agriculture Projects: An Institutional Theory Perspective. *Advanced Project Management*, 21(3), 7. https://doi.org/10.31235/osf.io/v4je2
- Dong, H., Dacre, N., Baxter, D., & Ceylan, S. (2022). *Understanding Agile in Project Management*. Association for Project Management. https://doi.org/10.61175/PUSU1455
- Dong, H., Dacre, N., Baxter, D., & Ceylan, S. (2024). What is agile project management? Developing a new definition following a systematic literature review. *Project Management Journal*, 21. https://doi.org/10.1177/87569728241254095
- Dudley, R., & Golden, D. (2021). The colonial pipeline ransomware hackers had a secret weapon: self-promoting cybersecurity firms. *ProPublica* (24 May 2021).
- Duho, K. C. T., Duho, D. M., & Forson, J. A. (2023). Impact of income diversification strategy on credit risk and market risk among microfinance institutions. *Journal of*

- *Economic and Administrative Sciences*, *39*(2), 523-546. https://doi.org/10.1108/JEAS-09-2020-0166
- Eggleton, D., Dacre, N., Cantone, B., & Gkogkidis, V. (2021). *Dynamic conditions for project success*. Association for Project Management. https://doi.org/10.61175/FXCU4654
- Eggleton, D., Dacre, N., Cantone, B., & Gkogkidis, V. (2023). From hypothesis to evidence: testing the Ika and Pinto four dimensional model of project success. *British Academy of Management*, 23. https://doi.org/10.2139/ssrn.5003846
- Eling, M., McShane, M., & Nguyen, T. (2021). Cyber risk management: History and future research directions. *Risk Management and Insurance Review*, 24(1), 93-125.
- Etikan, I., Musa, S. A., & Alkassim, R. S. (2016). Comparison of convenience sampling and purposive sampling. *American journal of theoretical and applied statistics*, 5(1), 1-4.
- Ezeoha, A. E. (2011). Banking consolidation, credit crisis and asset quality in a fragile banking system: Some evidence from Nigerian data. *Journal of financial regulation and compliance*, 19(1), 33-44.
- Falaiye, T., Elufioye, O. A., Awonuga, K. F., Ibeh, C. V., Olatoye, F. O., & Mhlongo, N. Z. (2024). Financial inclusion through technology: a review of trends in emerging markets. *International Journal of Management & Entrepreneurship Research*, 6(2), 368-379.
- Feinstein, Z., Halaj, G., & Sojmark, A. (2024). The not-so-hidden risks of hidden-to-maturity accounting: on depositor runs and bank resilience. *arXiv* preprint arXiv:2407.03285.
- Fernández-Villaverde, J., Guerrón-Quintana, P., Rubio-Ramirez, J. F., & Uribe, M. (2011). Risk matters: The real effects of volatility shocks. *American Economic Review*, 101(6), 2530-2561.
- Frait, J., & TomŠÍk, V. (2014). Impact and implementation challenges of the Basel Framework for emerging, developing and small economies. *Comparative Economic Studies*, 56, 493-516.
- Gambacorta, L. (2009). Monetary policy and the risk-taking channel. *BIS Quarterly Review December*.
- Ghosh, A. (2015). Banking-industry specific and regional economic determinants of non-performing loans: Evidence from US states. *Journal of Financial Stability*, 20, 93-104.
- Gkogkidis, V., & Dacre, N. (2020a). Co-Creating Educational Project Management Board Games to Enhance Student Engagement. *Game Based Learning*, 15. https://doi.org/10.48550/arXiv.2104.04063
- Gkogkidis, V., & Dacre, N. (2020b). Exploratory Learning Environments for Responsible Management Education Using Lego Serious Play. *ArXiv Computers and Society* 13. https://doi.org/10.48550/arXiv.2104.12539
- Gkogkidis, V., & Dacre, N. (2021). How to use Lego Serious Play as a Gamification Teaching and Learning Framework? A Responsible Management Approach. *SocArxiv*, 13. https://doi.org/10.31235/osf.io/vdzsu

- Gkogkidis, V., & Dacre, N. (2023). The educator's LSP journey: creating exploratory learning environments for responsible management education using Lego Serious Play. *Emerald Open Research*, *1*(12), 16. https://doi.org/10.1108/EOR-12-2023-0004
- Gong, Z., Dacre, N., & Dong, H. (2022). Fostering digital transformation through project integration management. *British Academy of Management*, 19. https://doi.org/10.2139/ssrn.5003601
- Gottschalk, R. (2014). *Institutional challenges for effective banking regulation and supervision in Sub-Saharan Africa*. ODI.
- Grima, S., Thalassinos, E., Cristea, M., Kadłubek, M., Maditinos, D., & Peiseniece, L. (2023). *Digital transformation, strategic resilience, cyber security and risk management*. Emerald Publishing Limited.
- Gubareva, M., & Borges, M. R. (2016a). Rethinking Framework of Integrated Interest Rate and Credit Risk Manag ement in Emerging Markets. In *Risk Management in Emerging Markets* (pp. 143-185). Emerald Group Publishing Limited. https://doi.org/10.1108/978-1-78635-452-520161017
- Gubareva, M., & Borges, M. R. (2016b). Rethinking framework of integrated interest rate and credit risk management in emerging markets. In *Risk Management in Emerging Markets: Issues, Framework, and Modeling* (pp. 143-185). Emerald Group Publishing Limited.
- Gubareva, M., & Borges, M. R. (2022). Governed by the cycle: interest rate sensitivity of emerging market co rporate debt. *Annals of Operations Research*, *313*(2), 991-1019. https://doi.org/10.1007/s10479-021-03972-x
- Hadiat, A. N. Q., & Anjani, N. P. N. Effective Risk Management Strategies for Small and Medium-Sized Enterp rises (SMEs) to Mitigate Interest Rate Fluctuation Risks and Ensure Fi nancial Sustainability. *International Journal of Business, Economics, and Social Development*, 5(2), 162-167. https://doi.org/10.46336/ijbesd.v5i2.657
- Hawkins, J., & Mihaljek, D. (2001). The banking industry in the emerging market economies: competition, consolidation and systemic stability: an overview. *BIS papers*, 4(4), 1-44.
- Hazaa, Y. M. H., Almaqtari, F. A., & Al-Swidi, A. (2021). Factors influencing crisis management: A systematic review and synthesis for future research. *Cogent Business & Management*, 8(1), 1878979.
- Hopper, T., Lassou, P., & Soobaroyen, T. (2017). Globalisation, accounting and developing countries. *Critical Perspectives on Accounting*, 43, 125-148.
- Hoskisson, R. E., Eden, L., Lau, C. M., & Wright, M. (2000). Strategy in emerging economies. *Academy of management journal*, 43(3), 249-267.
- Hsu, M.-w., Dacre, N., & Senyo, P. (2021a). Identifying Inter-Project Relationships with Recurrent Neural Networks: Towards an AI Framework of Project Success Prediction. *British Academy of Management*, 8. https://doi.org/10.2139/ssrn.3880328

- Hsu, M.-w., Dacre, N., & Senyo, P. K. (2021b). Applied Algorithmic Machine Learning for Intelligent Project Prediction: Towards an AI Framework of Project Success. *Advanced Project Management*, 21, 6. https://doi.org/10.31235/osf.io/6hfje
- Huo, X., Qian, Y., Siau, K. L., & Nah, F. F.-H. (2024). HCI in Business and Organizations: Digital Transformation with HCI, Metaverse, and AI Technologies. In *Human-Computer Interaction in Various Application Domains* (pp. 294-347). CRC Press.
- Jackson, P., & Perraudin, W. (2000). Regulatory implications of credit risk modelling. Journal of Banking & Finance, 24(1-2), 1-14.
- Jameaba, M.-S. (2024). Digitalization, emerging technologies, and financial stability: challenges and opportunities for the Indonesian banking sector and beyond. *Emerging Technologies, and Financial Stability: Challenges and Opportunities for the Indonesian Banking Sector and Beyond (April 26, 2024)*.
- John, K., Knyazeva, A., & Knyazeva, D. (2015). Employee rights and acquisitions. *Journal of Financial Economics*, 118(1), 49-69.
- Jorion, P. (2000). Risk management lessons from long-term capital management. *European financial management*, 6(3), 277-300.
- Kamasa, K., Ankrah, D., Tetteh, A., & Bonuedi, I. (2023). How Germane is Banking Regulation and Supervision to Financial Sector Stability in Emerging Economies? Empirical Evidence from Ghana. *African Journal of Economic Review*, 11(4), 66-84.
- Kamiya, S., Kang, J.-K., Kim, J., Milidonis, A., & Stulz, R. M. (2021). Risk management, firm reputation, and the impact of successful cyberattacks on target firms. *Journal of Financial Economics*, 139(3), 719-749.
- Kandasamy, J., Venkat, V., & Mani, R. S. (2023). Barriers to the adoption of digital technologies in a functional circular economy network. *Operations Management Research*, 16(3), 1541-1561.
- Kane, E. J. (1995). Difficulties of transferring risk-based capital requirements to developing countries. *Pacific-Basin Finance Journal*, 3(2-3), 193-216.
- Karanja, J. G., Karuti, J. K., & Senaji, T. A. (2022). Assessing the effect of credit risk control practices on lending performance of commercial banks in Kenya.
- Karim, M. Z. A., Chan, S.-G., & Hassan, S. (2010). Bank efficiency and non-performing loans: Evidence from Malaysia and Singapore. *Prague Economic Papers Issue 2, 2010, Pages 118-132*.
- Kasman, A., & Kasman, S. (2016). Bank size, competition and risk in the Turkish banking industry. *Empirica*, 43, 607-631.
- Kenny, C. (2018). The Equifax data breach and the resulting legal recourse. *Brook. J. Corp. Fin. & Com. L.*, 13, 215.
- Khan, S., Kabanov, I., Hua, Y., & Madnick, S. (2022). A systematic analysis of the capital one data breach: Critical lessons learned. *ACM Transactions on Privacy and Security*, 26(1), 1-29.
- Kharabsheh, B., & Gharaibeh, O. K. (2022). Determinants of banks' stability in Jordan. *Economies*, 10(12), 311.

- Kithinji, A. M. (2010). Credit risk management and profitability of commercial banks in Kenya.
- Kockum, F., & Dacre, N. (2021). Project Management Volume, Velocity, Variety: A Big Data Dynamics Approach. *Advanced Project Management*, 21, 6. https://doi.org/10.31235/osf.io/k3h9r
- Kou, G., Xu, Y., Peng, Y., Shen, F., Chen, Y., Chang, K., & Kou, S. (2021). Bankruptcy prediction for SMEs using transactional data and two-stage multiobjective feature selection. *Decision Support Systems*, *140*, 113429.
- Krishnamurthy, A. (2010). How debt markets have malfunctioned in the crisis. *Journal of Economic Perspectives*, 24(1), 3-28.
- Krumay, B., Bernroider, E. W., & Walser, R. (2018). Evaluation of cybersecurity management controls and metrics of critical infrastructures: A literature review considering the NIST cybersecurity framework. Secure IT Systems: 23rd Nordic Conference, NordSec 2018, Oslo, Norway, November 28-30, 2018, Proceedings 23,
- Kutum, I., & Al-Jaberi, K. (2015). Jordan banks financial soundness indicators. *International journal of finance & Banking Studies*, 4(3).
- Lane, P. R. (2012). The European sovereign debt crisis. *Journal of Economic Perspectives*, 26(3), 49-68.
- Levine, R. (1996). Foreign banks, financial development, and economic. *International financial markets: Harmonization versus competition*, 224.
- Louzis, D. P., Vouldis, A. T., & Metaxas, V. L. (2012). Macroeconomic and bank-specific determinants of non-performing loans in Greece: A comparative study of mortgage, business and consumer loan portfolios. *Journal of Banking and Finance*, *36*(4), 1012-1027. https://doi.org/10.1016/j.jbankfin.2011.10.012
- Maddaloni, A., & Peydró, J.-L. (2011). Bank risk-taking, securitization, supervision, and low interest rates: Evidence from the Euro-area and the US lending standards. *the review of financial studies*, 24(6), 2121-2165.
- Manda, V. K., & Rani, P. S. (2019). Crisis in the indian non-banking finance companies (NBFC) sector. *International Journal of Recent Technology and Engineering*, 8(4), 4502-4507.
- Manh, P. L., Dabscheck, D., Simpson, A., Temes, P., Shehhi, H. A., Balushi, Z. A., Jindong, Z., Li, Z., Jiang, F., Garro, F., Singh, D., Ojekhekpen, F., Guha, A., Montgomery, O., Metpally, F., Kapoor, V., Griffiths, M., Mathur, M., Aguiñaga, G. A., . . . Dacre, N. (2024). *Pulse of the Profession 2024 The Future of Project Work*. Project Management Institute. https://www.pmi.org/media/pmi/documents/public/pdf/learning/thought-leadership/pmi-pulse-of-the-profession-2024-report.pdf
- Mashhour, A., & Zaatreh, Z. (2008). A framework for evaluating the effectiveness of information systems at Jordan banks: An empirical study. *Journal of internet banking and commerce*, 13(1), 1-14.

- Mateev, M., Sahyouni, A., & Tariq, M. U. (2023). Bank regulation, ownership and risk taking behavior in the MENA region: policy implications for banks in emerging economies. *Review of Managerial Science*, 17(1), 287-338.
- Mateev, M., Usman Tariq, M., & Sahyouni, A. (2023). Efficiency, market concentration and bank performance during the COVID-19 outbreak: Evidence from the MENA region. *Plos one*, *18*(5), e0285403.
- Mazumder, M., & Sobhan, A. (2020). The spillover effect of the Bangladesh Bank cyber heist on banks' cyber risk disclosures in Bangladesh. *Journal of Operational Risk*, 15(4).
- Mazur, M. P. (2023). Economic growth and development in Jordan. Routledge.
- McLennan, M. (2022). The global risks report 2022 17th edition.
- Metrick, A. (2024). The failure of silicon valley bank and the panic of 2023. *Journal of Economic Perspectives*, 38(1), 133-152.
- Miller, M. J. (2003). Credit reporting systems and the international economy. Mit Press.
- Mitchener, K. J., & Trebesch, C. (2023). Sovereign debt in the twenty-first century. Journal of Economic Literature, 61(2), 565-623.
- Moges, H.-T., Dejaeger, K., Lemahieu, W., & Baesens, B. (2012). A total data quality management for credit risk: new insights and challenges. *International Journal of Information Quality*, *3*(1), 1-27.
- Moh'd AL-Tamimi, K. A., Jaradat, M. S., & Al-Rjoub, A. M. (2019). The Role of Central Bank of Jordan in Economic Development. *International Journal of Economics and Financial Research*, 5(10), 221-226.
- Mohamed, M., Al-Mhdawi, M. K. S., Dacre, N., Ojiako, U., Qazi, A., & Rahimian, F. (2024). Theoretical and Practical Instantiations of Generative AI in Construction Risk Management: An Analytical Exposition of Its Latent Benefits and Inherent Risks. Sage Advance, 14. https://doi.org/10.31124/advance.173139703.39850030/v1
- Mohsin, H. J., Hani, L. Y. B., Atta, A. A. B., Al-Alawnh, N. A. K., Ahmad, A. B., & Samara, H. H. (2023). The impact of digital financial technologies on the development of entrepreneurship: evidence from commercial banks in the emerging markets. *Corporate & Business Strategy Review*, 4(2), 304-312.
- Moussa, M. A. B. (2015). The determinants of bank liquidity: Case of Tunisia. *International journal of economics and financial issues*, 5(1), 249-259.
- Mpofu, T. R., & Nikolaidou, E. (2018). Determinants of credit risk in the banking system in Sub-Saharan Africa. *Review of development finance*, 8(2), 141-153.
- Nahar, J., Rahaman, M. A., Alauddin, M., & Rozony, F. Z. (2024). Big Data in Credit Risk Management: A Systematic Review Of Transformative Practices And Future Directions. *International Journal of Management Information Systems and Data Science*, 1(04), 68-79.
- Naili, M., & Lahrichi, Y. (2022). The determinants of banks' credit risk: Review of the literature and future research agenda. *International Journal of Finance & Economics*, 27(1), 334-360.

- Naimi-Sadigh, A., Asgari, T., & Rabiei, M. (2022). Digital transformation in the value chain disruption of banking services. *Journal of the Knowledge Economy*, 13(2), 1212-1242.
- Ng, A. W., & Kwok, B. K. (2017). Emergence of Fintech and cybersecurity in a global financial centre: Strategic approach by a regulator. *Journal of financial regulation and compliance*, 25(4), 422-434.
- Nikolaidou, E., & Vogiazas, S. (2017). Credit risk determinants in Sub-Saharan banking systems: Evidence from five countries and lessons learnt from Central East and South East European countries. *Review of development finance*, 7(1), 52-63.
- Nobanee, H., Ellili, N. O. D., Chakraborty, D., & Shanti, H. Z. (2024). Mapping the fintech revolution: how technology is transforming credit risk management. *Global Knowledge, Memory and Communication*.
- Nobanee, H., Hamill, P. A., Azmi, W., Chakraborty, D., & Nghiem, X.-H. (2023). In search of a safe haven in times of turbulence: Effects of First Republic Bank failure on global asset markets. *Heliyon*, *9*(10).
- Novaes Neto, N., Madnick, S., de Paula, M. G., & Malara Borges, N. (2020). A case study of the capital one data breach. *Stuart E. and Moraes G. de Paula, Anchises and Malara Borges, Natasha, A Case Study of the Capital One Data Breach (January 1, 2020).*
- Novak, A. N., & Vilceanu, M. O. (2019). "The internet is not pleased": twitter and the 2017 Equifax data breach. *The Communication Review*, 22(3), 196-221.
- Ofosu-Ampong, K. (2021). Determinants, barriers and strategies of digital transformation adoption in a developing country Covid-19 era. *Journal of Digital Science*, 3(2), 67-83.
- Ogunmokun, O. C., Mafimisebi, O. P., & Obembe, D. (2023). Prospect theory and bank credit risk decision-making behaviour: a systematic literature review and future research agenda. *SN Business & Economics*, *3*(4), 90.
- Ozili, P. K. (2024). Causes and consequences of the 2023 banking crisis. In *Governance and Policy Transformations in Central Banking* (pp. 84-98). IGI Global.
- Paligorova, T., & Sierra, J. (2012). Monetary policy and the risk-taking channel: Insights from the lending behaviour of banks. *Bank of Canada Review*, 2012(Autumn), 23-30.
- Peretti, K. K. (2008). Data breaches: What the underground world of carding reveals. Santa Clara Computer & High Tech. LJ, 25, 375.
- Pontin, D., & Dacre, N. (2024). The future of project management simulation exercises. *Engineering Education Research Network*, 14. https://doi.org/10.2139/ssrn.5003143
- Poudel, R. P. S. (2012). The impact of credit risk management on financial performance of commercial banks in Nepal. *International Journal of arts and commerce*, 1(5), 9-15.
- Prasad, E., Rogoff, K., Wei, S.-J., & Kose, M. A. (2003). Effects of financial globalization on developing countries: some empirical evidence. In *India's and China's recent experience with reform and growth* (pp. 201-228). Springer.

- Reinhart, C. M., & Rogoff, K. S. (2011). The forgotten history of domestic debt. *The economic journal*, 121(552), 319-350.
- Reynolds, D., & Dacre, N. (2019). Interdisciplinary Research Methodologies in Engineering Education Research. *Engineering Education Research Network*, 7. https://doi.org/10.48550/arXiv.2104.04062
- Saeed, M. (2024). The Influence of Transfer Pricing on International Tax Competition: A Case Study of Emerging Economies. *Social Dynamics Review*, 7(1).
- Saleh, I., & Abu Afifa, M. (2020). The effect of credit risk, liquidity risk and bank capital on bank profitability: Evidence from an emerging market. *Cogent Economics & Finance*, 8(1), 1814509.
- Saliba, C., Farmanesh, P., & Athari, S. A. (2023). Does country risk impact the banking sectors' non-performing loans? Evidence from BRICS emerging economies. *Financial Innovation*, 9(1), 86.
- Santomero, A. M. (1997). Commercial bank risk management: an analysis of the process. *Journal of Financial Services Research*, 12, 83-115.
- Saxena, S., & Villar, A. (2008). Hedging instruments in emerging market economies. *BIS* papers, 41.
- Schlackl, F., Link, N., & Hoehle, H. (2022). Antecedents and consequences of data breaches: A systematic review. *Information & Management*, 59(4), 103638.
- Scott, A. O., Amajuoyi, P., & Adeusi, K. B. (2024). Advanced risk management solutions for mitigating credit risk in financial operations. *Magna Scientia Advanced Research and Reviews*, 11(1), 212-223.
- Serkbayeva, Z., Zhumabayeva, M., Kassenova, G., Karimbayeva, G., Faizulayev, A., & Kulumbetova, D. (2024). Examining the impact of fintech and other factors on banking practices: QISMUT + 3 countries. *Cogent Business and Management*, 11(1), Article 2385069. https://doi.org/10.1080/23311975.2024.2385069
- Shen, L. (2014). The NIST cybersecurity framework: Overview and potential impacts. *Scitech Lawyer*, 10(4), 16.
- Siddique, A., Khan, M. A., & Khan, Z. (2022). The effect of credit risk management and bank-specific factors on the financial performance of the South Asian commercial banks. *Asian Journal of Accounting Research*, 7(2), 182-194.
- Simiyu, R. S. (2008). A survey of techniques of credit risk management in micro-finance institutions in Kenya University of Nairobi].
- Simpson, J. (2002). The impact of the Internet in banking: observations and evidence from developed and emerging markets. *Telematics and Informatics*, 19(4), 315-330.
- Singh, A. (1997). Financial liberalisation, stockmarkets and economic development. *The economic journal*, 107(442), 771-782.
- Sonjit, P., Dacre, N., & Baxter, D. (2021a). *Covid-19 & Homeworking Project Management Agility as the New Normal* British Academy of Management, Online, UK.
- Sonjit, P., Dacre, N., & Baxter, D. (2021b). Disruption and Agility Dynamics in Project Management Processes: An Institutional Theory Approach. *Advanced Project Management*, 21(7), 4. https://doi.org/10.2139/ssrn.3830762

- Sonjit, P., Dacre, N., & Baxter, D. (2021c). Homeworking Project Management & Agility as the New Normal in a Covid-19 World. *Advanced Project Management*, *21*(5), 5. https://doi.org/10.2139/ssrn.3823901
- Stamolampros, P., & Symitsi, E. (2022). Employee treatment, financial leverage, and bankruptcy risk: Evidence from high contact services. *International Journal of Hospitality Management*, 105, 103268.
- Stiglitz, J. E. (2003). Globalization and growth in emerging markets and the new economy. *Journal of policy modeling*, *25*(5), 505-524.
- Sultana, R. (2009). Mobile banking: Overview of regulatory framework in emerging markets. 4th Communication Policy Research, South Conference, Negombo, Sri Lanka,
- Tariq, E., Akour, I., Al-Shanableh, N., Alquqa, E., Alzboun, N., Al-Hawary, S., & Alshurideh, M. (2024). How cybersecurity influences fraud prevention: An empirical study on Jordanian commercial banks. *International Journal of Data and Network Science*, 8(1), 69-76.
- Thakor, A. V. (1996). The design of financial systems: An overview. *Journal of Banking & Finance*, 20(5), 917-948.
- Tite, C. N. J., Pontin, D., & Dacre, N. (2021a). Embedding Sustainability in Complex Projects: A Pedagogic Practice Simulation Approach. *Advanced Project Management*, 21, 7. https://doi.org/10.48550/arXiv.2104.04068
- Tite, C. N. J., Pontin, D., & Dacre, N. (2021b). Inspiring the Next Generation of Project Managers: Embedding Sustainability in Engineering Projects through Project Management Teaching and Learning. *Ingenium*, 9. https://doi.org/10.2139/ssrn.3880499
- Tosun, O. K. (2021). Cyber-attacks and stock market activity. *International Review of Financial Analysis*, 76, 101795.
- Tweneboah-Kodua, S., Atsu, F., & Buchanan, W. (2018). Impact of cyberattacks on stock performance: a comparative study. *Information & Computer Security*, 26(5), 637-652.
- Uddin, M. H., Ali, M. H., & Hassan, M. K. (2020). Cybersecurity hazards and financial system vulnerability: a synthesis of literature. *Risk Management*, 22(4), 239-309.
- Van Greuning, H., & Bratanovic, S. B. (2020). Analyzing banking risk: a framework for assessing corporate governance and risk management. World Bank Publications.
- Vong, J., & Song, I. (2015). Emerging technologies for emerging markets (Vol. 11). Springer.
- Wagdi, O., & Tarek, Y. (2022). The integration of big data and artificial neural networks for enhancing credit risk scoring in emerging markets: Evidence from Egypt. *International Journal of Economics and Finance*, 14(2), 32-43.
- Warren, M. (2024). The rise and fall of Silvergate Bank: lessons for prudential regulation of crypto-sector banking. *Journal of Banking Regulation*, 1-23.
- Wise, R. (2017). Reassessing the risks in emerging markets. *Journal of Risk Management in Financial Institutions*, 10(1), 74-77.

- Xu, K., Brito, M., Young, C., Al-Mhdawi, M. K. S., Dacre, N., & Baxter, D. (2023). Statistical analysis and modelling of small satellite reliability 33rd European Safety and Reliability Conference, Southampton, UK. https://doi.org/10.3850/978-981-18-8071-1 P627-cd
- Yanenkova, I., Nehoda, Y., Drobyazko, S., Zavhorodnii, A., & Berezovska, L. (2021). Modeling of bank credit risk management using the cost risk model. *Journal of Risk and Financial Management*, 14(5), 211.
- Youvan, D. C. (2024). Emergent Phenomena in Modern Financial Systems: Unanticipated Risks and Their Mitigation.
- Zalan, T., & Toufaily, E. (2017). The promise of fintech in emerging markets: Not as disruptive. *Contemporary Economics*, 11(4), 415.
- Zhang, Y., Dong, H., Baxter, D., & Dacre, N. (2023). Agile Meets Digital: A Systematic Literature Review on the Interplay between Agile Project Management and Digital Transformation *British Academy of Management*, 16. https://doi.org/10.2139/ssrn.5007866
- Zou, Y., Mhaidli, A. H., McCall, A., & Schaub, F. (2018). "I've Got Nothing to Lose": Consumers' Risk Perceptions and Protective Actions after the Equifax Data Breach. Fourteenth Symposium on Usable Privacy and Security (SOUPS 2018),